Manuscript Rejection Notification

Dear [Author's Name],

Thank you for submitting your manuscript titled "[Manuscript Title]" to [Journal Name]. After careful consideration and review, we regret to inform you that we cannot accept your manuscript for publication.

Reviewer Comments Summary:

- **Reviewer 1:** The methodology lacks clarity and requires more detailed explanation.
- **Reviewer 2:** The results section is insufficiently supported by data and lacks necessary statistical analysis.
- **Reviewer 3:** The novelty of the findings is questioned, as similar studies have been previously published.

We encourage you to consider the feedback provided by the reviewers and make the necessary revisions if you decide to submit your manuscript elsewhere.

Thank you once again for your interest in [Journal Name]. We wish you the best in your future research endeavors.

Sincerely, [Editor's Name] [Journal Name]